Author’s response to reviews

Title: A pilot study of the S-MAP (Solutions for Medication Adherence Problems) intervention for older adults prescribed polypharmacy in primary care: Study protocol

Authors:
Deborah Patton (deborah.patton@qub.ac.uk)
Jill Francis (Jill.Francis.1@city.ac.uk)
Ella Clark (ellavclark@gmail.com)
Felicity Smith (f.j.smith@ucl.ac.uk)
Cathal Cadogan (cathalcadogan@rcsi.ie)
Cristin Ryan (cristin.ryan@tcd.ie)
Carmel Hughes (c.hughes@qub.ac.uk)

Version: 1 Date: 05 Aug 2019

Author’s response to reviews:
Letter has been uploaded with revised paper